

Working Papers Number 12/31

Early Lessons from Country Experience with Matching Contribution Schemes for Pensions

Economic Analysis Madrid, December 2012



Early Lessons from Country Experience with Matching Contribution Schemes for Pensions*

Robert Holzmann^a, Richard Hinz^b and David Tuesta^c

December 2012

Abstract

Matching defined contribution schemes are gaining popularity in both rich and poor countries as a promising means to reduce gaps in the participation in formal pension systems. Matching contributions by employers, the government, or both to defined contribution schemes are used alone or jointly with other interventions to motivate participation in pension schemes. Although it remains far too early to develop firm conclusions or policy guidance, this chapter provides an overview of the currently available evidence that is presented in this volume and offers preliminary observations about the potential use of this design. This experience, mostly derived from higher-income countries that is now being supplemented with some early experience from other settings, suggests that matching is moderately effective in increasing program participation but not generally measurably effective in raising contributions and thus benefit levels. Other interventions—which are increasingly guided by lessons from behavioral economics and finance—may prove to be more effective and typically cost much less, which may help explain some of the differences in outcomes across countries. It is not yet clear how transferrable the experience in higher-income environments will be to other settings; considerable further evaluation is needed before any firm conclusions can be reached.

Keywords: pensions, defined contributions, matching contributions, pension coverage.

JEL: G23, H55, J32.

^{*} This paper has been previously published as the Chapter 1 of the book titled "Matching Defined Contributions for Pensions: A Review of International Experience" edited by Richard Hinz, Robert Holzmann, David Tuesta and Noriyuki Takayama, World Bank 2012. The chapters mentioned in the present document are those of the referred book. This paper was presented at review meetings and seminars in Madrid and Washington, D.C., and has profited from written comments by internal and external reviewers, in particular by Will Price, Rafael Rofman, and Sylvester Schieber. The authors are grateful for the comments and suggestions received but are ultimately responsible for any gaps or errors. The views expressed are their own and do not necessarily reflect those of the institutions with which they are affiliated.

a: World Bank and University of Malaya

b: World Bank

c: BBVA Research

1. Addressing the Coverage Gap

Achieving broad pension coverage and adequate levels of income protection in old age remains an elusive goal for nearly every country. In general, coverage has advanced with development and growth in income, and there is a strong relationship between the level of per capita income and participation in formal pension systems. However, there are considerable differences among countries at similar levels of development in pension coverage and in the way in which participation in pension systems has evolved in different settings. The differences in experiences and outcomes indicate that context, the design of the system, and the path of its development play a central role in the dynamics of pension coverage and benefit levels.

In the vast majority of countries, less than half the working population is currently covered by a formal pension scheme. In low-income and developing countries, the number of working-age adults participating in a pension scheme is very often less than 1 in 10. Many middle-income countries have seen coverage rates decline in recent decades despite the expansion of coverage mandates and efforts to reform their systems to establish stronger individual incentives (Rofman and Oliveri 2012). In higher-income countries, coverage levels under mandated schemes remain high, but fiscal pressures caused by the generous benefits provided to early cohorts and exacerbated by rapidly falling fertility rates are now imposing the need for reductions in future benefit levels. Such reductions will require a substantial increase in supplementary retirement savings if income replacement rates are to be maintained.

All of these factors have brought the imperative for the extension of coverage to the forefront of the pension policy debate. An earlier World Bank publication investigated the role of social pensions and other transfers to increase retirement income support across countries of different development levels (Holzmann, Robalino, and Takayama 2009). This volume extends the consideration of how to provide adequate income to elderly populations by focusing on the potential role of matching contributions to induce broader participation in pensions and other retirement saving to help close the coverage and adequacy gap.

1.1 The Potential Promise of Matching Contribution Systems

An approach that has been adopted in a growing number of high-income countries is the provision of matching contributions. These contributions provide more tangible incentives for individuals to participate in pension funds than the more traditional approach of mandating participation and providing preferential tax treatment, especially for low income groups and individuals who may not participate in the formal labor force and therefore receive no advantage from tax-based incentives.

In principle, matching contributions may be provided for public programs or by the sponsors of private occupational plans and could be associated with either defined contribution or defined benefit systems. In practice, there are numerous cases of both public and private schemes utilizing the design; however, nearly all current examples are associated with various types of individual retirement savings accounts. This volume, which reviews experience to date and seeks to derive some initial observations and policy lessons, focuses on what is here termed matching defined contribution (MDC) schemes.

In all of these systems, the matching design feature has the common goal of increasing system participation and saving levels. Four characteristics are common to MDC schemes examined here: individual accounts, defined contribution, direct contribution from sponsor to complement individual contributions, and prefunding of benefits. The prevalence of defined contribution systems is likely a reflection of two factors. First, although in theory it is possible to incorporate a contribution match in a defined benefit system, the linkages between the match and the benefit received are complex and less transparent. More importantly, the populations these arrangements are seeking to reach, especially lower-income groups or those in developing countries, may have irregular earning patterns over their life cycle or be predominantly engaged in the informal sector. The systems reviewed vary considerably in the structure and level of the matching contribution. Although the majority of schemes provide ex ante matches, some provide ex post matches.

The use of individual accounts with defined contributions as the underlying structure provides direct linkages between density and level of contributions to align individual incentives and provides transparency in the value of benefits. Directly matching contributions provides an immediate and easily understandable value proposition to prospective entrants to the system. Funding of the accumulated contributions and returns with financial institutions should offer credibility, portability, and appropriate returns. The expected saving incentives created by MDCs are an alternative or complement to other potential incentives such as preferential tax treatment and the presentation of choices in a way intended to influence behavior (what Thaler and Sunstein have termed "nudging"), as well as efforts to create a more conducive old-age saving environment such as financial education and straightforward advocacy. MDC schemes are also attractive because they define and constrain future fiscal exposure. They should increase coverage but not

encourage informality. If individuals are sufficiently incentivized by the match, the accumulated amounts at retirement may reduce or even eliminate the need for basic benefits. Inducing funded supplemental coverage will facilitate reductions in mandated public systems for higher-income groups. Ideally, coverage should increase and the level of labor informality decrease at lower fiscal costs than expanding non contributory systems or subsidizing earnings-based defined benefit programs.

1.2 Policy Questions

Addressing the coverage gap raises a range of interrelated policy questions that vary by setting. In low- and middle-income countries, the primary challenge is to expand pension coverage beyond civil servants and the small proportion of the workforce employed in the formal sector. When much of the labor force has no fixed employer, or is self-employed, the traditional method of expanding coverage through wage-based mandated contributions is not a viable option. In these countries, establishing pension systems with incentives for participation that are attractive to low-income and young people with no prior experience with social insurance and saving systems imposes enormous challenges. These groups nearly always struggle to meet short-term needs and require liquidity in any savings that they are able to accrue in order to manage a variety of risks. Effective solutions must not impose disincentives for participation in the formal sector through high payroll tax contributions or create adverse redistribution through tax-based subsidies that provide value only to the highest-income groups.

In other middle-income settings, the challenge is to maintain the coverage rates achieved in earlier decades in the face of increasing informality of the workforce due to transitions from centrally planned economies or changes in labor patterns resulting from economic development and competition in a global economy. In middle-income and most higher-income countries, the imperative is also to establish retirement income and saving systems that are able to supplement the diminished capacity of earlier earnings-based public systems in order to provide adequate income replacement for future cohorts. The social policy concern is accentuated by low, stagnating, and at times falling coverage of old-age pensions and other social programs—a stark contrast with expectations that emerging economies would follow the same path as the current high-income economies, achieving coverage expansion in step with income growth.

When initially confronted with this coverage challenge, policy makers believed that reforms designed to establish better links between contributions and benefits in mandated and earnings-related schemes, often by creating individual and funded accounts, would overcome these problems. To date, limited success has been achieved, however, and in some cases coverage has declined following such reforms.

Closely related to these social policy issues are broader economic concerns over the high and often rising level of informality. This growth is perceived as hampering economic development because workers in the informal sector are considered less prone to learn, innovate, and use productivity-enhancing technologies. Applying basic coverage options—such as universal or meanstest benefits for the elderly—to take care of informal sector workers may prove counterproductive, because such options reduce the incentive to become formal while increasing the pressure on formal sector workers to become informal as their tax burden increases. A number of recent studies indicate that this may be happening in countries across Latin America (see, for example, Aterido, Hallward-Driemeier, and Pagés 2011; Levy 2008; and Ribe, Robalino, and Walker 2012) although at present the empirical evidence remains tentative.

In all settings, fiscal concerns arise from the cost of coverage expansion through non contributory basic schemes, the need to control the fiscal costs of traditional national earnings-based defined benefit systems, and the effects of sustained informality on productivity and public revenues.

All three issues are potentially linked in a downward spiral in which (1) coverage concerns lead to the introduction or strengthening of basic provisions, which (2) constrain employment and increase informality, (3) increasing pressure to leave the formal sector, (4) worsening the fiscal position of the public pension scheme, (5) leading to benefit cuts, which in turn increase the need for better basic benefits. Against this background, policy options in low- and middle-income countries are limited and largely untested. The most obvious and direct approach is to establish stronger incentives for participation in formal pension and saving schemes. The key challenge is to develop a design that will motivate lower-income groups—which require powerful, immediate, and readily understood incentives to overcome their inherent consumption preference and liquidity constraints—to direct their limited resources to retirement income. Such a system must also be attractive to informal sector workers, many of whom may be of moderate or even higher-income levels, while not increasing the incentives for workers to leave the formal sector.

1.3 Purpose and Organization of this Volume¹

Several high-income countries, most notably Germany, New Zealand, and the United States, have adopted the MDC design to complement benefit levels under public schemes. Other countries, including Japan and the United Kingdom, have recently initiated MDC schemes to raise savings earmarked for retirement income. This has fostered interest in the design in a variety of other settings. Emerging economies in Asia (China, India, Thailand) and Latin America (Chile, Colombia, Mexico, Peru) have implemented or are considering implementing MDC-based schemes to encourage participation in voluntary and sometimes mandated schemes by individuals who would otherwise have no coverage at all. Such matching incentives may not in principle be restricted to defined contribution schemes, but examples of their use in defined benefit systems are rare (only the case of a new system in Republic of Korea is included in this volume) but worthwhile to review.

Despite the growing experience, there is little consolidated knowledge to provide evidence-based policy guidance about the role and limits of MDCs for expanding retirement savings, about best practice in the design and implementation of MDC programs, or about the interaction of MDC policies with other interventions such as financial literacy as complementary (or substitute) approaches. Nearly all of the experience with these arrangements and associated research has come from higher-income countries. Applicability to the vastly different circumstances in middle-income and developing countries remains uncertain.

Against the backdrop of pilots in emerging and developed economies, the Social Protection Unit of the World Bank's Human Development Network, in cooperation with the Research Institute for Policies on Pension and Aging (RIPPA) of Japan, organized a conference in June 2011 to provide a forum for sharing information and analyzing experience from around the world with this emerging design. The Spanish Bank BBVA, which manages pension funds in a number of Latin American countries, participated in the conference and subsequently joined with the World Bank and RIPPA to provide the resources to supplement and organize the material presented at the conference and produce this publication.

This volume—based on the presentations at the conference—provides overviews and analyses intended to inform the ongoing design and use of MDC schemes. The publication thus presents a first stocktaking of country experiences and some limited observations about the potential role and effectiveness of the design that can be derived from experience to date. It is not an effort to formulate or articulate a World Bank policy position or to provide any specific direction to countries considering the design. The evidence is far too limited to support such an effort, and as noted above, there is insufficient evidence to assess the transferability of the more extensive experience from higher-income countries to other settings. Not considered at the conference and in the volume are similar matching design approaches to increased coverage under health care or other social insurance programs. There is, however, considerable value in consolidating the knowledge that can be gleaned from the wide range of experiments with the design, considering what lessons are beginning to emerge and how these can inform future initiatives and a research agenda; these issues are summarized in this introductory chapter.

The book contains four parts. Part I provides an overview of the more general issues and experience in expanding supplementary pension coverage in Organisation for Economic Cooperation and Development (OECD) countries to establish a broader framework of the challenges of coverage expansion. Part II reviews experience with MDCs in high-income countries. Part III describes early efforts in lower- and middle-income settings. Part IV provides an overview of lessons from the emerging field of behavioural economics and reviews key issues in the enabling environment and the main parameters likely to be relevant in establishing an MDC in a developing country context.

The remainder of this introductory chapter is organized as follows. The next section offers a conceptual framework for the objectives, intervention, mechanisms, and modalities of MDCs. The following sections provide country examples, extract some tentative lessons from their experience, and draw preliminary policy conclusions.

-

¹ We refer to the book of Hinz et al (2012) in this paper. All chapters to which we refer from here.

2. Objectives, Interventions, Mechanisms, and Modalities

Assessing the effectiveness of a policy intervention starts with consideration of the objectives, a clear understanding of how the core elements of the intervention are defined, and how it expects to achieve the desired outcomes. Such a "theory of change" is critical to assess the effectiveness of any intervention and is at the core of monitoring and evaluation efforts²

2.1 Setting Objectives

Policy discussions in countries that have or are planning to introduce MDC systems suggest three primary objectives: expanding pension coverage, reducing informality, and increasing fiscal efficiency. Considering these requires some degree of conjecture as countries are rarely explicit about the objectives of a policy intervention and even less specific regarding how to measure outcomes. Furthermore, the political discussion is often overloaded with secondary objectives driven by group interests or political imperatives.

2.1.1 Expanding Coverage

The increase in basic or supplementary benefit coverage is probably the primary objective in most countries that have or are planning to introduce MDC-type pension systems. This objective is particularly relevant for low- and middle-income countries, where most people are not afforded even basic coverage for old-age income and health care.

Supplementary coverage is additional coverage for people who participate in mandated systems (as workers or beneficiaries) but whose benefit levels are considered inadequate. Providing such coverage is particularly relevant in high-income countries, where the large majority of the population is covered under earnings-related schemes or everyone is covered under universal schemes that only partially replace their income in old age.

One measure of the success of the basic benefit coverage is the number of people who receive a benefit at or above a minimum level (typically the poverty line). For supplementary benefits, success in coverage expansion may be measured by the number of people whose benefits rise above a threshold of a specified percentage or amount. In this case, however, the benefit increase may be caused by shifts in savings from unsubsidized to subsidized forms without an increase in net wealth

Obviously, the success of the program rarely depends entirely on the financial incentives provided but will also be affected by government and operational capacities in addition to other factors.

2.1.2 Reducing Informality

Reducing the incentives to participate in the informal sector (that is, evading the costs of participating in mandated social insurance schemes) is another important objective of an MDC system, particularly in middle-income countries. This objective may be conceptualized as establishing conditions that will encourage individuals to make contributions and acquire rights for the first time or contribute more to an MDC scheme. Success can be measured in terms of the number of registered participants as well as by changes in the contribution density of participants. MDCs should also not create incentives to reduce the level of pension savings by those already contributing.

In developing countries, mandated participation in pension systems is often conditional on the number of employees in a firm; workers in firms below a certain size may be exempt from mandatory contributions. In other instances, workers may not be able to contribute because the firm is not formally registered or licensed. In such cases, an MDC alone may not be the appropriate intervention.

2.1.3 Increasing Fiscal Efficiency

MDCs create fiscal costs—either directly, through the matching contributions by governments, or indirectly, through preferential tax treatment of individuals or their employers. A simple measure of success would be the increase in the coverage rate or adequacy target per currency unit of public

² All modern monitoring and evaluation books apply a similar logic, although they may use different terminology. For recent highly readable publications on monitoring and evaluation,see Gertler and others (2010); Khandker, Koolwal, and Samad (2009); and Leeuw and Vaessen (2009). Also see the websites of the Massachusetts Institute of Technology's Poverty Action Lab (http://www.povertyactionlab.org/), the World Bank (http://www.worldbank.org/oed/ecd), and the World Bank's Strategic Impact Evaluation Fund (http://go.worldbank.org/X81HJAZSGO).

expenditure for matching and/or foregone tax revenue resulting from the increased savings that is excluded from taxation.

Another fiscal element that is often explicitly part of the design of an MDC scheme is the cost saving through reduced transfers to the elderly in the form of universal or means-tested benefits. These kinds of ex post transfers may include the costs for minimum benefits in an earnings-related scheme. Very optimistically, one could imagine a take-up of low-level matching contributions that largely eliminates the need for these transfers, with the saved fiscal costs well exceeding the new fiscal costs of the matching payments. Very pessimistically, one could imagine that MDCs merely shift savings from unsubsidized to subsidized forms—or, even worse, that most people substitute public funds for individual savings, with take-up concentrated largely in the upper-income strata.

2.1.4 Other Objectives

In a system with progressive income tax rates, favorable tax treatment of pension contribution disproportionately favors people with higher incomes. By providing a subsidy that is directly proportional to the contribution, an MDC scheme may not suffer from this shortcoming unless, again, the take-up is concentrated among high-income earners.

MDCs may also be linked with the objective of facilitating the transition to a fully or partially funded system by inducing individual contributions to support a funded or capitalized reserve from which benefits will be paid.

2.2 Defining the Intervention: Core Elements of Matching Defined Contribution Schemes

The following features define an MDC scheme:

- **Individual account**. Accounting records should be maintained to clearly distinguish individual contribution and retirement saving outcomes.
- **Defined contributions**. Benefits are solely based on the accrued value of contributions and earnings on accumulated assets.
- **Sponsor**. Employers, the government, or other sponsoring entities make direct financial contributions to encourage individuals' participation.
- **Own contributions**. Regular and own contributions by individuals are expected, although sometimes after an initial "kick-start" contribution.
- **Funding**. In principle, schemes can be funded or unfunded (using a derived crediting rate and notional accounts). In practice, there are no examples of matching being incorporated into unfunded retirement savings schemes.
- **Mandatory or voluntary**. Mandatory schemes typically focus on extension of coverage. Voluntary schemes are typically employer based or designed to supplement other systems with broad coverage but relatively low benefit levels.

Other known design elements of MDCs are complementary and potentially substitutive to matching. As with matching, these intend to encourage the participation of individuals in retirement saving programs. The main examples include the following:

- Tax preferences for contributions and/or benefits under a comprehensive income tax approach
 that aims to eliminate distortions of taxation on savings and move toward a consumption-type
 structure (exempting contributions and interest from taxation but taxing the disbursement, or
 taxing contributions but leaving interest earnings and disbursement tax free)
- Nudging or choice architecture (Thaler and Sunstein 2009) to motivate individuals to participate
 in savings plans; specific mechanisms include automatic enrolment, default contribution levels,
 investment options, and thresholds for matching levels
- Financial education and related interventions to create the enabling environment for individuals to learn about the importance of planning and retirement saving and offer them support to acquire the requisite skills, attitudes, and behaviors.

2.3 Envisaged Mechanism: The Theory of Change

Identifying the mechanism through which outcomes will be achieved begins with an analysis of the problems to be solved and the rationale for government intervention. Public interventions are traditionally undertaken for two key reasons: to correct market failures and to redistribute income. Market failures are often linked to asymmetric information, which leads to poorly functioning (or nonexistent) markets. Government interventions are intended to substitute for or improve market outcomes. Redistribution is undertaken to correct perceived failures in the way markets generate and distribute income. To these motivations for public intervention, a third has been added in recent years: to correct the behavioral limitations of individuals. A fourth reason may be to correct government failure itself—by, for example, redesigning social insurance programs that do not achieve sufficient coverage due to poor design or implementation.

Individuals may rationally evade government social insurance programs for a variety of reasons, including the high costs associated with participation in the formal sector, liquidity preferences, the poor fit between a program and individual preferences, and lack of trust. These factors can be conceptualized as increasing the discount rate individuals apply to pension schemes. An MDC scheme seeks to increase the internal rate of return of a pension scheme in order to increase participation, contributions, or both.

Individuals may fail to employ market-based social risk management instruments to address long-term contingencies, such as old age, for a variety of reasons, including the absence of appropriate instruments, the inability to plan ahead, and the existence of more immediate shocks for which risk management instruments are not available. In the absence of an additional external motivation, these factors may make it rational for some people to focus on the short rather than the long term. Under such circumstances, matching may actually distort rational individual decisions and divert resources from, say, human capital accumulation to retirement saving.

Whether contributions grow as a result of the provision of a match is a priori undetermined. A saving subsidy in the form of a match potentially creates both a substitution effect (which makes current consumption more expensive and hence increases current saving) and an income effect (which increases the demand for current and future consumption and hence decreases current saving). MDC interventions with multitier structures have even more complicated effects on individuals, which depend on the position of the individual before and after the introduction of or change to the intervention. Some configurations are predicted to clearly encourage or discourage saving; the effect of others is indeterminate. The predictions are broadly borne out in the country lessons discussed in the chapters.

2.4 Design and Administrative Modalities

The main design features of MDC schemes include the following:

- **Matching rates** typically range from 25 percent to 100 percent but can reach 300 percent and more. For supplementary MDCs offered by employers, there may be complex structures with lower match rates for higher contribution levels or no match for a first tier before a declining match applies. For basic matches offered by governments, flat-rate and multi-tiered matching contributions (which address informal sector workers, for whom earnings cannot be easily established) are more typical.
- **Thresholds and ceilings** on the contribution base (linked to a multiple of average income) or overall limits on matched amounts direct subsidies to lower-income groups and limit fiscal costs.
- **Eligibility conditions** for matching include very specific characteristics of the beneficiary (income level, age, family status, number and age of children, employment status, company size, level of formality) to focus the match on a target population.
- **Matching contributions** provide an ex ante transfer linked with ex post transfer (minimum pension) that is conditional on a required amount of prior contributions (length or level) by the beneficiary or family members.
- **Withdrawals** can be made to purchase a first home or to buffer periods of unemployment or other contingencies.
- In addition to the match, consumption-type tax treatment is provided for savings, subject only to income tax at time of receipt (similar to other pension savings plans).
- **Payout modalities** include lump sums, phased withdrawals, mandated immediate or deferred annuitization, and annuitization defaults (for example, minimum annuity levels, deferred annuities).

Administrative modalities include the following:

- Contribution collection by employer, financial institution, social security institution, or local "aggregators" (nongovernmental organizations or others that collect payments from contributors)
- Recordkeeping and client communication by financial institution or regional or national social security fund
- Asset management by pension/health funds, specialized asset manager, or regional or national social security institution
- Benefit disbursement by financial institution or regional or national social security institution.

3. Country Examples

MDC schemes reflect diverse policy objectives at different stages of the pension system development. Common features can be organized by the main coverage extension scenarios:

- Supplementing universal (basic or means-tested) benefits
- Supplementing low or reduced earnings-related benefits
- Expanding coverage within the mandated social insurance scheme
- Expanding coverage outside the mandated social insurance scheme (universal approach)
- Expanding coverage outside the mandated social insurance scheme (sector- or group- specific approach).

This section illustrates these coverage objectives with country examples detailed in later chapters in this volume to distil lessons and to raise policy and research questions.

3.1. Supplementing Universal (Basic or Means-Tested) Benefits

Two countries have sought to use MDC schemes to supplement universal benefits or limit mandated provisions: New Zealand (which introduced such a plan in 2007) and the United Kingdom (which had intended to introduce such a plan in 2010 but cancelled the program following the election of a new government).

New Zealand is one of the few countries with an old-age program in which everyone receives a pension (this type of program is known as a *demogrant*). Every resident of New Zealand age 65 and older who has lived in the country for at least 10 years receives a flat rate pension based on a certain percentage of the average wage. Provision for old age above this basic and taxable benefit was left to voluntary savings, creating issues of adequacy of retirement income.

After discussions and a defeated referendum on introducing mandated earnings related benefits to eventually replace the demogrant, the government introduced an MDC-type scheme, the design of which was, to a significant degree, informed by the emerging field of behavioral economics; this design is both complex and comprehensive. The KiwiSaver program, described in chapter 5, utilizes several tiers of matching-type incentives. It provides a flat contribution (known as a kickstart) on the opening of an account, a tax credit (actually a pure subsidy) for contributions, and mandates matching contributions by employers (2 percent of payroll, to increase to 3 percent by 2013). Employees can choose to contribute 2 percent, 4 percent, or 8 percent of their earnings. It combines these subsidies with an auto-enrolment feature in which new employees are automatically enrolled in the system but afforded the ability to opt out after a specified period. It allows preretirement withdrawals for several purposes, most notably the purchase of a first home.

The United Kingdom has had a Beveridge-inspired basic pension scheme with flat-rate contributions and benefits since the 1940s. The country has a long history of trying to improve saving outcomes of all types—from short term (precautionary) saving to very long-term pension and retirement income saving—in order to complement basic pension plan provisions. The Saving Gateway program, described in chapter 6, planned to use matching contributions to increase the savings of people with low incomes. Programs— tested through limited-scale pilot efforts in 2002–04 and 2005–07—experimented with different matching rates, contribution limits, eligibility rules, and recruitment mechanisms.

In 2008, the Labour government then in power decided to roll out Saving Gateway nationally to up to 8 million people, or 20 percent of the population between the ages of 16 and 65. In May 2010, the newly elected Conservative-Liberal Democrat coalition government cancelled the scheme as part of its broader program of fiscal retrenchment. However, with the National

Employment Savings Trust (NEST), the U.K. government is aiming to increase coverage of the privately funded pension system from 2012 by using auto-enrolment together with matching as another example of nudging behavior. A minimum contribution requirement is gradually being introduced, ultimately reaching 4 percent of wages by individuals, 3 percent from employers and 1 percent from the government in tax relief. Employees and employers are free to contribute more, and some employers will match higher contributions. Auto-enrolment is being used because such employer matching has been in existence for many years, and many employees did not join schemes even when very generous matches were offered by employers.³

3.2. Supplementing Low or Reduced Earnings-Related Benefits

A number of MDC schemes are related to low levels of income replacement or reforms that reduced the generosity of public earnings-related schemes, primarily in high-income OECD countries. There are wide variations in design and operation across countries.

A number of high-income countries have introduced compensatory supplementary and funded pensions with more modest amounts of direct fiscal support through flat-rate and similar subsidies (beyond consumption-type tax treatment). To provide some background on the overall issue of providing supplementary coverage, chapter 2 reviews the experience among OECD member countries with policies that encourage private pension saving.

The 401(k) plans used in the United States (named after a section of the tax code that authorized the particular type of tax-preferred saving arrangement), examined in chapter 3, are probably the most important and most investigated MDC scheme in the world. These plans emerged in the early 1980s as part of the employer-sponsored pension system that supplements the relatively low (on average, less than 40 percent) income replacement rates of the mandatory public social security system. These were part of the transition in voluntary employer pension coverage from defined benefit to defined contribution plans. The underlying arrangement enables workers to determine the level of pre-tax contributions (technically known as salary deferrals) to a defined contribution plan. To ensure that the value of this tax preference did not disproportionately favor individuals with higher incomes (who not only had a greater ability to save but who, because of the progressive income tax system, obtained a larger value from the tax deferral), rules were established that limited the amount higher-income workers could contribute; this amount is linked to the overall average share of income deferred by all participants. This rule led employers to create a wide range of matching contribution designs to induce higher contributions from low- and average-income workers to enable the higher-paid to take full advantage of potential tax preferences. The widespread use of 401(k) plans and the diversity of matching arrangements have been the subject of extensive study and are the source of much of the knowledge about the behavioral effects of various matching designs that are discussed in chapter 15.

In 2001, Germany introduced another important variant of the supplemental arrangement that is known as the Riester pension (in recognition of the former minister of labor and social security who was a main proponent of the initiative) after having imposed a significant prospective reduction in the value of public pension system benefits. The Riester pension plans that were introduced in stages from 2002 to 2008 involve a means-tested match of contributions from the government, an additional per child subsidy, a tax preference on contributions up to a maximum level, and an associated (largely annuitized) payout plan. The development of and experience with this system is discussed in chapter 4. This MDC scheme is heavily subsidized by the budget and at times substitutes for corporate pensions. After a slow start and several design changes, including simplifications, in 2005, Riester pension plans took off very quickly. Saving incentives have been effective in reaching households with children; they have been somewhat less successful in attracting low-income earners.

In more recent years, Japan has, like the United States, sought to expand its occupational pension system to include defined contribution plans as it attempts to address dramatic demographic changes that will constrain the public social security system and thereby relieve some of the pressures on traditional corporate defined benefit plans. The Japanese pension system and the new defined contribution plans are discussed in chapter 7. In an interesting variation, matching was introduced in 2011 with a design in which the employee is permitted to match the employer's contribution to the defined contribution plan—reversing the typical arrangement. Thus far, however, there is no evidence that this design will meet with any more success than other defined contribution plans, which have had only very limited acceptance, thus indicating the importance of context and incentives in matching arrangements.

^{3.2.} See http://www.direct.gov.uk/en/Pensionsandretirementplanning/Companyandpersonalpensions/ WorkplacePensions/DG_200722 for on overview of the government's plans and time schedule.

3.3. Expanding Coverag e within the Mandated Social Insurance Scheme

Social insurance schemes in middle-income countries have difficulty expanding coverage to low-income groups. Some vulnerable groups, such as youth, are difficult to reach, but their early integration into a social insurance scheme is important for later behavior. Everywhere in the world, self-employed workers exhibit low participation and contribution efforts, particularly in rural areas.

A number of countries are attempting to encourage participation by groups that are difficult to integrate by offering matching incentives within the mandated social insurance scheme. The match is financed by the budget or through redistributed contributions.

Korea's social insurance pension scheme, discussed in chapter 8, was established in 1988 and made universal in 1999. It covers all working-age (age 18-60) adults who make or are exempted from making contributions. Contribution delinquency remains an issue, creating concerns for future pension adequacy. To strengthen participation of farmers and fishers, the government has, since 1995, when the national pension extended its compulsory coverage to all rural residents, offered contribution subsidies of 50 percent of the total contribution (with a cap). This subsidy is scheduled to end in 2013. Although the Korean pension scheme is of the defined benefit type and does not squarely fall into the MDC definition, it offers one of the few matching schemes in middle/high-income countries that have been evaluated. As the subsidy does not apply to other self-employed groups beyond farmers and fishers, it offers a natural experiment for testing its effectiveness. It is found to have had a moderate effect in increasing participation in the system by individuals who would otherwise be expected to evade making contributions.

In Colombia—which is examined along with Mexico and Peru in chapter 10—at least three MDC schemes have been established to encourage voluntary contributions. Two are already in operation, one as part of the funded individual account system and one as part of the alternative unfunded defined benefit scheme. Individuals must choose between the two schemes. In both cases, the match provides minimum income guarantees for retirees and is financed by contribution income.

Mexico has had at least two matching-type schemes in operation since the mandated individual account pension scheme was established in the late 1990s. The first scheme targets low-wage workers by providing a flat-rate "social contribution" to all participants below an income ceiling (introduced in 2009) equal to 5.5 percent of the presumed minimum wage for each day of work. A second scheme, for civil servants, was introduced with the 2007 reform that moved their pensions toward a funded defined contribution scheme. The match—a government match of Mex\$3.25 for each Mex\$1 of employee contribution, with a ceiling of 2.0 percent of the contribution base for the employee and 6.5 percent for the employer—should increase contributions.

Peru's MDC scheme, originally legislated in 2008, was designed to promote coverage for workers in small and microenterprises while enhancing competitiveness and encouraging participation in the formal labor market. The matching component of this law, the Welfare Pension System (Sistema de Pensiones Sociales), has recently been included in the 2012 reform of the private pension system. This scheme is focused only on microenterprise employees (those working in firms with no more than 10 employees) and is mandatory for people under age 40 and earning less than 1.5 times the legal minimum wage. Both the contribution rate and the government matching are to be defined during the year. It is likely that the government will finally implement this scheme.

Chile, examined in chapter 9, introduced two youth employment subsidy schemes, with the objective of promoting formal youth employment through incentives for both the supply of and demand for labor. The schemes' introduction, in 2008, occurred around the time of a major pension reform that introduced ex post subsidies—in particular, guaranteed old-age income through the solidarity pillar of the pension system. The Subsidio Previsional a los Trabajadores Jóvenes (SPTJ) scheme provides an explicit subsidy for social security contributions. A first component (introduced in October 2008) amounts to 50 percent of social security contributions at minimum wage. This, however, is paid to the employer to provide a subsidy for the cost of hiring younger workers while also providing an incentive for contributions to the social security system. A second component (introduced in July 2011) provides a matching payment to the worker of the same amount to subsidize contributions.

3.4. Expanding Coverage outside the Mandated Social Insurance Scheme (Universal Approach)

In most low- and many middle-income countries, the majority of workers work in the informal sector. Integration of the workers into the formal sector pension scheme is unlikely to be realistic in the near term. A few countries, including India and Thailand, do offer voluntary coverage outside the mandated social insurance scheme and provide a government match to induce participation.

In India, discussed in chapter 12, less than 10 percent of the population works in the formal sector—and much of the formal sector employment is in the public sector. To address a looming pension problem in an existing defined benefit system for central government workers, the reform of 2004 introduced a funded defined contribution scheme for new entrants to the civil service and unbundled recordkeeping and asset management. With an administrative structure established through this reform, the New Pension Scheme provided the potential infrastructure for an MDC scheme for all informal sector workers. An effort to expand the system was initiated in 2010, by which an annual matching contribution of Re 1,000 (about \$20) was offered for all workers who enrol and pay contributions of Re 1,000–12,000 (about \$20 to \$225) a year, with no means test applied. To enhance the decision architecture, the scheme uses "aggregators" at the village level to collect contributions and has a simplified account structure with lower fees. The scheme's very recent implementation makes it impossible to draw conclusions about its success. An early look at a small set of data indicates that participation in the scheme may be associated with income and education levels and might be negatively associated wit access to alternative sources of retirement savings.

Thailand, discussed in chapter 14, initiated a national MDC scheme for informal sector workers in 2012. Individual deposits can be made at any time, with a minimum deposit of B 50 (about \$2). The government match and ceiling are graduated by age, with a 50 percent match for people age 15–30 up to a maximum of B 3,000 (about \$100), an 80 percent match for people age 31–50 with a maximum of B 4,800 (about \$155), and a 100 percent match for people age 51–60 with a limit of B 6,000 (about \$200).

These efforts in India and Thailand trigger many questions that will require further evidence and rigorous evaluation to answer, including the following:

- What is the appropriate matching structure for informal sector workers in low and middleincome countries? In particular, how important are matching rates compared with contribution ceilings?
- What is the role of matching compared with other participation determinants, such as the decision architecture and potential members' perceptions of the service providers?
- How effective are aggregators and efforts to increase access through "points of presence" (the establishment of a means to make contributions in a village such as a bank or post office) in increasing participation into rural areas?
- What mechanisms encourage continued pension saving efforts beyond increased participation?

3.5. Expanding coverage outside the Mandated Social Insurance Scheme (Sector- and Group-Specific Approaches)

In countries with a large informal sector and the desire to increase formal sector participation, a universal voluntary system for informal sector workers may not be the best way to achieve higher levels of pension coverage. A focus on specific groups may be justified, however. Self-employed workers lend themselves to such an approach, as they are difficult to integrate into a formal mandated scheme.

A number of countries have started to move in this direction. China, discussed in chapter 11, started a voluntary MDC pilot for the rural sector in 2009; the program was expanded to the urban sector in 2011, and full national coverage is envisaged by 2013. These voluntary schemes will coexist with the mandated urban pension scheme, which covers about half the urban workforce.

China's introduction of the National Rural Pension Scheme and the Urban Resident Pension Scheme has been one of the most ambitious voluntary pension saving and minimum elderly assistance schemes in a low- or middle-income country. Both schemes have innovative features. They provide a basic pension benefit from age 60 on if a vesting period of 15 years is fulfilled. Individuals select a contribution level of between Y 100-500 (\$16-\$80) per year. A partial match is then provided by local governments, with a minimum required match of Y 30 (\$5), although this may be at a higher level that is locally determined. The rapid expansion of these voluntary schemes that now are reported to cover more than 350 million participants may be linked to the minimum pension benefit, which is offered immediately if conditions are fulfilled. But it also reflects

the influence the government has on inducing participation in public social insurance systems as well as a solid advocacy campaign.

Another important challenge in developing countries is reaching independent workers, who offer their labor in often irregular patterns—such as fishers in offering their labor and skills to the owners of boats. Because this employment model does not lend itself to long-term relationships, these kinds of workers cannot make steady contribution payments into a social insurance fund. Their situation calls for innovative new payment and financing solutions adapted to the particular circumstances of the targeted workers that include nudging elements as well as matching-type contributions by contractors. An exploratory study of Cape Verde and Tunisia, presented in chapter 13, offers an outside-of-the-box thought piece on this issue.

These country examples of innovative MDC schemes suggest that coverage can be expanded by moving beyond simple matching design to improving the decision architecture. Linking ex post and ex ante transfers, as China has done, or offering new financing and payment structures, as proposed for Cape Verde and Tunisia, exemplify this approach.

4. Tentative Lessons from the Experience with Matching Contributions

The country experience and reported results offer a rich, although incomplete (and likely somewhat biased), body of evidence on the use of matching contribution arrangements. Most of the rigorous evaluation of the dynamics and outcomes of matching are focused on participation and contribution effects in 401(k) plans. These address particular groups (generally higher-income employers in a high-income country who are offered the chance to participate in an employer-sponsored plan) and therefore may not be relevant for other countries. These studies do not address some of the key questions for other settings such as impact on informality or overall fiscal effectiveness, as the matching is by employers and any fiscal effects are indirect. Most other country experiences have not (yet) been subject to rigorous evaluation; in many circumstances, there has been no evaluation at all. Consequently, the discussion presented below includes more hypotheses than lessons that have been inspired by empirical results and validated in other countries

4.1. Expanding Coverage

4.1.1. Participation

There is consistent empirical support across country income levels that matching is effective in increasing participation. The evidence from a few high-income countries (mostly the United States) indicates positive but modest effects of matching on participation, with overall effects increasing participation in the range of 5–10 percent of potential beneficiaries. The associated finding that a 25 percent match of individual contributions is associated with about a 5 percent increase in participation appears to be robust across a range of programs and analysis in the United States. This magnitude is also broadly consistent with results from Korea, where a 50 percent match for farmers and fishers increased the probability of making a pension contribution by 7.4 percent. The presence of a large initial match—a significant element of the KiwiSaver system in New Zealand—elicited enrolment from many people with little or no earnings, including children, providing further evidence of the potential effectiveness of significant matches on at least initial enrolment.

4.1.2. Increase in Retirement Saving

The effect of the match on the saving rate is typically found to be small, and the sign is not always positive or statistically significant. This finding is consistent with the theoretical ambiguity arising from the conflict between substitution and income effects. What seems to emerge consistently is that the structure of the match—the matching rate, thresholds, and caps—does have significant consequences. Consistent across estimates (essentially from the United States and, to a lesser extent, the United Kingdom), the match threshold seems to have a greater impact than the matching rate. Providing a lower match of 25 percent on contributions for a higher level of 10 percent of pay will induce individuals to save more than a higher match of 50 percent for a lower level of 5 percent of pay, although both formulations result in similar costs to the organization providing the match (see chapters 3, 6, and 15). A possible explanation for this result is that matching acts as a signalling device or implicit advice on saving levels. Also notable is the "stickiness" of saving levels, as evidenced by the fact that most people remain at contribution levels that were established as defaults even when the defaults are subsequently reduced.

4.1.3. Increase in Overall Saving/Total Pension Wealth

Comprehensive evidence from the United States on the effect of scheme saving on other saving is mixed, but suggests up to 20–30 percent net increases in saving levels (see chapters 3 and 15). In the United Kingdom (chapter 6), the saving rate increased but the measured net worth of individuals did not change. Evidence from Germany (chapter 4) suggests that matched saving did not squeeze out other saving.

4.1.4. Other Determinants for Participation and Contribution Efforts

The evidence from developed and developing countries strongly suggests that other features of savings programs and related interventions may have a critical—perhaps even a dominant—effect on participation and contribution levels. Most of these features have not yet been subject to the rigorous testing across a range of settings that could begin to distinguish between inherent effects and those associated with a particular set of circumstances, cultural setting, or population group.

- Automatic enrollment and defaults. Evidence from the United States, the United Kingdom, and New Zealand suggests that making participation the default option has two to four times the impact of the match. (Of course, an automatic enrollment default option works only under formal employment conditions.) The role of other default schemes on contribution efforts is mixed and at times negative, possibly because of inertia or the low default contribution rate.
- Simplification of design and access. Empirical results for the United States and the United Kingdom and lessons from the German Riester pensions suggest that simplified design affects participation and, perhaps, contribution/saving efforts.
- Social marketing and advocacy. Retirement saving remains an objective that most will embrace but find difficult to implement. In the United States, employers have found that information sessions and advocacy are a useful adjunct to the incentives of matching contributions. In Germany, take-up of the match increased with greater awareness of the scheme associated with information campaigns. New Zealand has coupled introduction of its system with information and advocacy campaigns which are perceived to have had the expected effect. There is some very preliminary evidence of a positive impact through the introduction of account aggregators in India. Yet rigorous empirical evidence is missing on the effectiveness of information campaigns for short-term participation and long term contribution efforts.

4.2. Reducing Informality

Evidence on the effect of MDC schemes on informality is very limited and mixed. There is no empirical evidence that the modest matching schemes in Colombia or Mexico have reduced informality or increased coverage, nor any clear expectation that the soon-to-be-implemented matching scheme in Peru will lead to these outcomes. This is possibly because of the small size of the programs and important distortions in the labor market. Data from national household surveys in these three South American countries show an enormous potential for saving in the informal sectors. The estimate for Korea suggests that the match for farmers and fishers had a modest positive impact on participation in the national pension scheme. The matching programs in Chile to incentivize the participation of young workers in the formal labor market—and hence in the pension scheme— increased participation, but the programs have not yet been subject to rigorous evaluation. The rural pilots in China that started in 2009 reportedly reached 358 million rural workers as of the end of 2011, and full coverage (of some 500 million people) is envisaged in 2013. The pilots increased coverage but, strictly speaking, had no effect on informality. This is likely to be because the target group had very little potential to become formal sector workers to begin with.

A number of countries (including China, India, and Thailand) have established or are planning voluntary matching programs in parallel to formal matched or unmatched schemes. Individuals joining these voluntary programs have no obligations to join the mandated scheme (as they do in Germany and the United States). Not enough evidence is available to determine whether these schemes create disincentives to formalization.

4.3. Increasing Fiscal Efficiency

Assessing the effect of matching schemes on coverage and informality is relatively straightforward, and effectiveness comparisons can be done within or across similar schemes. To measure fiscal efficiency requires the pricing of the intervention and comparison with alternatives or a counterfactual. This analysis is hardly ever done, however, as these schemes are designed and implemented—making consideration of potential fiscal effects more art than science.

In considering the possible fiscal efficiency of matching schemes, two comparisons are proposed: (1) a comparison of the fiscal costs with the additional savings volume created by the match and (2) a comparison of fiscal costs for the match with the costs of alternative interventions such as ex post subsidies. The most useful comparison may depend on the purpose of the intervention. If the objective is to promote supplementary coverage, evaluating the marginal increase in savings seems more relevant. If promoting basic coverage is the objective, the more relevant comparison is with alternative interventions.

4.3.1. Comparison to Additional Savings Created

Comparison of total fiscal costs with MDC savings created can be done on a flow and stock basis. Each requires some heroic assumptions.

- Comparing the annual contributions by participants to annual fiscal costs. Assuming that all contributions are new saving, fiscal effectiveness requires that the ratio of the annual flow of new savings to the annual cost of matches (a fiscal efficiency ratio) be larger than 1 so that the public expenditure and potential public dis-saving are at least compensated by additional contribution revenues of equal magnitude. Taking account of distortions (for example, through changes in general revenue collections) would increase the opportunity costs and therefore increase the required fiscal efficiency ratio. Using empirical results of new savings created of, say, only a third, fiscal effectiveness would require an efficiency ratio greater than 3. For the German Riester pensions, for example, the share of annual contributions to direct fiscal costs is slightly above 2 and falling (chapter 4).
- Comparing the additional national capital stock created with the accumulated fiscal expenditure. The idea behind this comparison is that on a net basis (aggregate new savings less aggregate cost of the subsidy), the match should increase the pension wealth of the elderly and the capital stock on which future benefits are paid by more than the overall cost of the subsidy. Such a calculation would take account of compensating or strengthening effects. Ex ante projections with an appropriate overlapping generations model would offer first indications of effects; for the more relevant ex post evaluation, the data and estimation requirements need to be developed.

4.3.2 Comparison to Ex Post Subsidy

Are matching contributions (ex ante subsidies) for voluntary or mandated schemes less expensive than non-contributory or subsidized benefit levels (ex post subsidies)? Measuring fiscal efficiency requires considering the likely cost and comparing it with alternatives. Doing so yields the following considerations:

- A demogrant provides everyone with a minimum transfer in old age, regardless of individual circumstances. This approach is very effective in distributive terms, but not fiscally efficient because the leakages are high as many receiving the benefit do not need it to maintain the level of consumption in old age that such a benefit is intended to achieve. The same minimum income support for needy elderly can be generated at much lower cost when targeted transfers are provided in the form of general social assistance or categorical social pensions (Grosh and Leite 2009). Of course, targeting, however well done, will lead to inclusion and exclusion errors.
- Compared to both demogrants and targeted ex post benefits, an MDC scheme can be constructed that is fiscally more efficient as long as individuals do some additional saving and targeting works effectively. If individuals are not induced by the match to increase their saving, for whatever reason, then matching leads to distributively inferior results. This has implications for the lowest-income groups where saving capacity for old age is limited. If individuals do some saving but ex ante targeting does not work well, it can lead to either fiscally less efficient outcomes than ex post targeting or distributionally less efficient outcomes or both. This will depend on the size of the inclusion and exclusion errors and the reaction of individuals to the subsidy provided. Unfortunately, the relevant experience and data are not available to undertake such a comparison.

None of the countries reviewed appears to have established a comprehensive set of outcome measures or undertaken estimates of the fiscal effectiveness of the matching design in relation to alternatives. No studies have been undertaken to provide comparative measures even at the conceptual level of the cost-effectiveness of key parameters. Such a study should be relatively simple at the individual level; however, at the macro level, the comparison would have to make a number of heroic assumptions to allow for comparability.

5. Preliminary Policy Conclusions and Next Steps

Because few MDC schemes have been subjected to a rigorous impact evaluation, policy lessons are by necessity very preliminary. Based on the evidence presented in this volume, a few conclusions and suggestions for future analysis can nevertheless be drawn:

- 1. Empirical evidence, collected largely from high-income countries, suggests that MDC schemes raise participation in pension systems moderately and have at best modest (and in some instances ambiguous) effects on contribution levels of participating individuals. Both findings are consistent with theoretical predictions. The persistence of any of these effects—and ultimately their influence on lifetime wealth accumulation and the provision of retirement income—will require much longer-term assessment. The net impact of matching on individual wealth and macroeconomic savings levels is empirically even more difficult to assess and subject to conflicting views. The weight of the limited evidence to date for high-income and perhaps some high-middle-income countries is that MDCs will make a helpful but insufficient contribution to solving the challenges of pension coverage, income maintenance, and informality of labor. They may, however, be effective for special groups that are difficult to reach by other means.
- 2. It is unclear to what extent the results for high-income countries will translate into the context of lower- and middle-income countries. The groups for which there is experience in high-income countries may not be representative of the broader population, nor are they likely to share characteristics and behavior patterns with target populations in low- and middle-income settings, which are likely to be characterized by many more constraints in their everyday life and much larger shocks for which social risk management instruments are not available. As a result, the incentives by MDC programs to enhance retirement savings may translate less well or at least differently. Since few programs of this kind existed until recently in developing economies, and none has been subject to rigorous impact evaluation, not much is yet known about how the experience in other settings will translate; caution is therefore advised.
- 3. The threshold and other parameters of the match seem to have as much, or even a greater, effect as the overall level of the match on behavior and saving outcomes in high-income countries. Similar empirical evidence is not yet available for low and middle-income countries. In view of the many more binding constraints in developing countries, complex matching rates and other parameters may be required to achieve the envisaged outcomes. A key design challenge in these settings will be to balance this with a need for simplicity and transparency. At least equally important for participation and saving effects are interventions directed toward overcoming behavioral and other limitations, in particular through the provision of information and financial education, a choice architecture consistent with the observed inertia in behavior, and social marketing, public service announcements, and other types of advocacy. Key considerations in designing such programs include the following:
 - Information about the subsidized scheme and its operation is critical in creating awareness, which seems to be an important factor in changing behavior (in general and for retirement income saving in particular).
 - Many (but not all) experts believe that the ability to understand the offered saving products and to apply that knowledge is a critical factor for participation in social risk management programs (whether mandated or voluntary, unsubsidized or matched).
 - The setting of defaults and other forms of choice architecture are increasingly recognized as key determinants of outcomes. Policy makers need to take advantage of lessons from behavioral economics about harnessing the power of inertia (in particular through automatic enrolment with limited opt-outs), simplifying administrative processes, and setting parameters.
 - Advocacy and education efforts—such as seminars, public service advertising, social marketing, publicly sponsored retirement savings information websites, and entertainment education—may be very important in promoting MDC schemes.
- 4. Individuals at all income levels evade participation in mandated schemes for a variety of reasons, including ineffective design choices, poor alignment of incentives with the environmental context, and lack of trust in public and financial institutions. Addressing these issues may go a long way toward increasing participation. These efforts should be undertaken before matching schemes are introduced.
- 5. The effect that pension-related matching incentives may have in changing employment patterns or reducing the level of informality in developing and transition economies remains wholly unknown. Thus far, there is little evidence that these schemes are sufficient to change individual decisions. This remains a difficult issue to effectively evaluate, but one that merits investigation despite the methodological challenges.

- 6. Examples and a framework are lacking for a comprehensive cost-benefit analysis for MDC designs to compare them with potential alternatives. Rigorous impact evaluation focused on microeconomic assessment is necessary to guide program design. This needs to be complemented by a more comprehensive welfare analysis, including cost-benefit evaluation of alternative approaches. Such analysis should address areas not covered in the research to date including consideration of the following:
 - A pure saving instrument (and pooling risks over time by individuals) or a traditional risk management instrument (pooling risks across individuals) in relation to matching designs
 - Differentiation of MDC outcomes in relation to various income groups; theoretical
 considerations and empirical indications suggest that matching to improve retirement saving
 is not likely to be effective for the poorest in society, and initial evidence suggests large
 substitution effects among higher-income groups
 - Gender analysis of MDC, addressing differences in labor force participation longevity, and other factors likely to create different outcomes
 - The longer-term role of matching designs and possible development of exit strategies for their use as a transitional device to generate participation as economies meet thresholds where coverage expansions become feasible.
- 7. Finally, it will be important to consider the possible future role and new designs of MDC schemes under different social policy scenarios, such as the following:
 - The marginalization or even the demise of pay-as-you-go earnings-based (Bismarckian) public pensions in countries with a high level of informality
 - A further reduction of the mandatory defined contribution pension systems in countries with low informality but also facing demographically driven reductions in the generosity of public earnings-related and basic schemes
 - The need for more portability of social benefits and the portability of the subsidy element of MDCs across borders.

References

Aterido, Reyes, Mary Hallward-Driemeier, and Carmen Pagés. 2011. "Does Expanding Health Insurance beyond Formal-Sector Workers Encourage Informality? Measuring the Impact of Mexico's Seguro Popular." IZA Discussion Paper 5996, Institute for the Study of Labor, Bonn.

Gertler, Paul J., Sebastian Martinez, Patrick Premand, Laura B. Rawlings, and Christel M. J. Vermeersch. 2010. Impact Evaluation in Practice. Washington, DC: World Bank.

Grosh, Margaret, and Phillippe Leite. 2009. "Defining Eligibility for Social Pensions: A View from a Social Assistance Perspective." In Closing the Coverage Gap: The Role of Social Pensions and Other Retirement Income Transfers, ed. R. Holzmann, D. A. Robalino, and N. Takayama, 161–86. Washington, DC: World Bank.

Holzmann, R., D. A. Robalino, and N. Takayama. 2009. Closing the Coverage Gap: The Role of Social Pensions and Other Retirement Income Transfers. Washington, DC: World Bank.

Hinz, R., R. Holzmann, D. Tuesta y N. Takayama. 2012. Editors. Matching Contributions for Pensiones: A Review of International Experience. Washington, DC. World Bank.

Khandker, S. R., G. B. Koolwal, and H. Samad. 2009. Handbook on Quantitative Methods of Program Evaluation. Washington, DC: World Bank.

Leeuw, F., and J. Vaessen, 2009. Impact Evaluations and Development: NONIE Guidance on Impact Evaluation. Washington, DC: World Bank.

Levy, Santiago. 2008. Good Intentions, Bad Outcomes. Washington, DC: Brookings Institution.

Ribe, Helena, David Robalino, and Ian Walker. 2012. From Right to Reality: Incentives, Labor Markets, and the Challenge of Achieving Universal Social Protection in Latin America and the Caribbean. Latin American Development Forum Series. Washington, DC: World Bank.

Rofman, Rafel, and Maria Laura Oliveri. 2012. "La cobertura de los sistemas previsionales en América Latina: conceptos e indicadores." (English version in preparation.) Series de Documentos de Trabajo sobre Políticas Sociales No. 7, World Bank, Buenos Aires.

Thaler, Richard H., and Cass R. Sunstein. 2009. Nudge: Improving Decisions about Health, Wealth, and Happiness. New York: Penguin Books.



Working Papers

09/01 **K.C. Fung, Alicia García-Herrero and Alan Siu:** Production Sharing in Latin America and East Asia.

09/02 **Alicia García-Herrero, Jacob Gyntelberg and Andrea Tesei:** The Asian crisis: what did local stock markets expect?

09/03 **Alicia García-Herrero and Santiago Fernández de Lis:** The Spanish Approach: Dynamic Provisioning and other Tools.

09/04 **Tatiana Alonso:** Potencial futuro de la oferta mundial de petróleo: un análisis de las principales fuentes de incertidumbre.

09/05 **Tatiana Alonso:** Main sources of uncertainty in formulating potential growth scenarios for oil supply.

09/06 **Ángel de la Fuente y Rafael Doménech:** Convergencia real y envejecimiento: retos y propuestas.

09/07 **KC FUNG, Alicia García-Herrero and Alan Siu:** Developing Countries and the World Trade Organization: A Foreign Influence Approach.

09/08 **Alicia García-Herrero, Philip Woolbridge and Doo Yong Yang:** Why don't Asians invest in Asia? The determinants of cross-border portfolio holdings.

09/09 **Alicia García-Herrero**, **Sergio Gavilá and Daniel Santabárbara**: What explains the low profitability of Chinese Banks?

09/10 **J.E. Boscá, R. Doménech and J. Ferri:** Tax Reforms and Labour-market Performance: An Evaluation for Spain using REMS.

09/11 **R. Doménech and Angel Melguizo:** Projecting Pension Expenditures in Spain: On Uncertainty, Communication and Transparency.

09/12 **J.E. Boscá, R. Doménech and J. Ferri:** Search, Nash Bargaining and Rule of Thumb Consumers.

09/13 **Angel Melguizo, Angel Muñoz, David Tuesta y Joaquín Vial:** Reforma de las pensiones y política fiscal: algunas lecciones de Chile.

09/14 **Máximo Camacho:** MICA-BBVA: A factor model of economic and financial indicators for short-term GDP forecasting.

09/15 **Angel Melguizo, Angel Muñoz, David Tuesta and Joaquín Vial:** Pension reform and fiscal policy: some lessons from Chile.

09/16 Alicia García-Herrero and Tuuli Koivu: China's Exchange Rate Policy and Asian Trade.

09/17 **Alicia García-Herrero, K.C. Fung and Francis Ng:** Foreign Direct Investment in Cross-Border Infrastructure Projects.

09/18 **Alicia García Herrero y Daniel Santabárbara García:** Una valoración de la reforma del sistema bancario de China.

09/19 C. **Fung, Alicia García-Herrero and Alan Siu:** A Comparative Empirical Examination of Outward Direct Investment from Four Asian Economies: China, Japan, Republic of Korea and Taiwan.

09/20 Javier Alonso, Jasmina Bjeletic, Carlos Herrera, Soledad Hormazábal, Ivonne Ordóñez, Carolina Romero y David Tuesta: Un balance de la inversión de los fondos de pensiones en infraestructura: la experiencia en Latinoamérica.

09/21 Javier Alonso, Jasmina Bjeletic, Carlos Herrera, Soledad Hormazábal, Ivonne Ordóñez, Carolina Romero y David Tuesta: Proyecciones del impacto de los fondos de pensiones en la inversión en infraestructura y el crecimiento en Latinoamérica.



- 10/01 **Carlos Herrera:** Rentabilidad de largo plazo y tasas de reemplazo en el Sistema de Pensiones de México.
- 10/02 Javier Alonso, Jasmina Bjeletic, Carlos Herrera, Soledad Hormazabal, Ivonne Ordóñez, Carolina Romero, David Tuesta and Alfonso Ugarte: Projections of the Impact of Pension Funds on Investment in Infrastructure and Growth in Latin America.
- 10/03 Javier Alonso, Jasmina Bjeletic, Carlos Herrera, Soledad Hormazabal, Ivonne Ordóñez, Carolina Romero, David Tuesta and Alfonso Ugarte: A balance of Pension Fund Infrastructure Investments: The Experience in Latin America.
- 10/04 **Mónica Correa-López y Ana Cristina Mingorance-Arnáiz:** Demografía, Mercado de Trabajo y Tecnología: el Patrón de Crecimiento de Cataluña, 1978-2018.
- 10/05 **Soledad Hormazabal D.:** Gobierno Corporativo y Administradoras de Fondos de Pensiones (AFP). El caso chileno.
- 10/06 **Soledad Hormazabal D.:** Corporate Governance and Pension Fund Administrators: The Chilean Case.
- 10/07 **Rafael Doménech y Juan Ramón García:** ¿Cómo Conseguir que Crezcan la Productividad y el Empleo, y Disminuya el Desequilibrio Exterior?
- 10/08 **Markus Brückner and Antonio Ciccone:** International Commodity Prices, Growth, and the Outbreak of Civil War in Sub-Saharan Africa.
- 10/09 **Antonio Ciccone and Marek Jarocinski:** Determinants of Economic Growth: Will Data Tell?
- 10/10 **Antonio Ciccone and Markus Brückner:** Rain and the Democratic Window of Opportunity.
- 10/11 **Eduardo Fuentes:** Incentivando la cotización voluntaria de los trabajadores independientes a los fondos de pensiones: una aproximación a partir del caso de Chile.
- 10/12 **Eduardo Fuentes:** Creating incentives for voluntary contributions to pension funds by independent workers: A primer based on the case of Chile.
- 10/13 **J. Andrés, J.E. Boscá, R. Doménech and J. Ferri:** Job Creation in Spain: Productivity Growth. Labour Market Reforms or both.
- 10/14 Alicia García-Herrero: Dynamic Provisioning: Some lessons from existing experiences.
- 10/15 **Arnoldo López Marmolejo and Fabrizio López-Gallo Dey:** Public and Private Liquidity Providers.
- 10/16 Soledad Zignago: Determinantes del comercio internacional en tiempos de crisis.
- 10/17 **Angel de la Fuente and José Emilio Boscá:** EU cohesion aid to Spain: a data set Part I: 2000-06 planning period.
- 10/18 Angel de la Fuente: Infrastructures and productivity: an updated survey.
- 10/19 **Jasmina Bjeletic, Carlos Herrera, David Tuesta y Javier Alonso:** Simulaciones de rentabilidades en la industria de pensiones privadas en el Perú.
- 10/20 **Jasmina Bjeletic, Carlos Herrera, David Tuesta and Javier Alonso:** Return Simulations in the Private Pensions Industry in Peru.
- 10/21 **Máximo Camacho and Rafael Doménech:** MICA-BBVA: A Factor Model of Economic and Financial Indicators for Short-term GDP Forecasting.
- 10/22 **Enestor Dos Santos and Soledad Zignago:** The impact of the emergence of China on Brazilian international trade.
- 10/23 **Javier Alonso, Jasmina Bjeletic y David Tuesta:** Elementos que justifican una comisión por saldo administrado en la industria de pensiones privadas en el Perú.



- 10/24 **Javier Alonso, Jasmina Bjeletic y David Tuesta:** Reasons to justify fees on assets in the Peruvian private pension sector.
- 10/25 **Mónica Correa-López, Agustín García Serrador and Cristina Mingorance-Arnáiz:**Product Market Competition and Inflation Dynamics: Evidence from a Panel of OECD Countries.
- 10/26 **Carlos A. Herrera:** Long-term returns and replacement rates in Mexico's pension system.
- 10/27 Soledad Hormazábal: Multifondos en el Sistema de Pensiones en Chile.
- 10/28 Soledad Hormazábal: Multi-funds in the Chilean Pension System.
- 10/29 **Javier Alonso, Carlos Herrera, María Claudia Llanes y David Tuesta:** Simulations of long-term returns and replacement rates in the Colombian pension system.
- 10/30 **Javier Alonso, Carlos Herrera, María Claudia Llanes y David Tuesta:** Simulaciones de rentabilidades de largo plazo y tasas de reemplazo en el sistema de pensiones de Colombia.
- 11/01 Alicia García Herrero: Hong Kong as international banking center: present and future.
- 11/02 **Arnoldo López-Marmolejo:** Effects of a Free Trade Agreement on the Exchange Rate Pass-Through to Import Prices.
- 11/03 Angel de la Fuente: Human capital and productivity
- 11/04 **Adolfo Albo y Juan Luis Ordaz Díaz:** Los determinantes de la migración y factores de la expulsión de la migración mexicana hacia el exterior, evidencia municipal.
- 11/05 **Adolfo Albo y Juan Luis Ordaz Díaz:** La Migración Mexicana hacia los Estados Unidos: Una breve radiografía.
- 11/06 **Adolfo Albo y Juan Luis Ordaz Díaz:** El Impacto de las Redes Sociales en los Ingresos de los Mexicanos en EEUU.
- 11/07 María Abascal, Luis Carranza, Mayte Ledo y Arnoldo López Marmolejo: Impacto de la Regulación Financiera sobre Países Emergentes.
- 11/08 María Abascal, Luis Carranza, Mayte Ledo and Arnoldo López Marmolejo: Impact of Financial Regulation on Emerging Countries.
- 11/09 **Angel de la Fuente y Rafael Doménech:** El impacto sobre el gasto de la reforma de las pensiones: una primera estimación.
- 11/10 **Juan Yermo:** El papel ineludible de las pensiones privadas en los sistemas de ingresos de jubilación.
- 11/11 Juan Yermo: The unavoidable role of private pensions in retirement income systems.
- 11/12 **Angel de la Fuente and Rafael Doménech:** The impact of Spanish pension reform on expenditure: A quick estimate.
- 11/13 **Jaime Martínez-Martín:** General Equilibrium Long-Run Determinants for Spanish FDI: A Spatial Panel Data Approach.
- 11/14 David Tuesta: Una revisión de los sistemas de pensiones en Latinoamérica.
- 11/15 David Tuesta: A review of the pension systems in Latin America.
- 11/16 **Adolfo Albo y Juan Luis Ordaz Díaz:** La Migración en Arizona y los efectos de la Nueva Ley "SB-1070".
- 11/17 **Adolfo Albo y Juan Luis Ordaz Díaz:** Los efectos económicos de la Migración en el país de destino. Los beneficios de la migración mexicana para Estados Unidos.
- 11/18 **Angel de la Fuente:** A simple model of aggregate pension expenditure.
- 11/19 Angel de la Fuente y José E. Boscá: Gasto educativo por regiones y niveles en 2005.
- 11/20 **Máximo Camacho and Agustín García Serrador:** The Euro-Sting revisited: PMI versus ESI to obtain euro area GDP forecasts.



- 11/21 Eduardo Fuentes Corripio: Longevity Risk in Latin America.
- 11/22 Eduardo Fuentes Corripio: El riesgo de longevidad en Latinoamérica.
- 11/23 **Javier Alonso, Rafael Doménech y David Tuesta:** Sistemas Públicos de Pensiones y la Crisis Fiscal en la Zona Euro. Enseñanzas para América Latina.
- 11/24 **Javier Alonso, Rafael Doménech y David Tuesta:** Public Pension Systems and the Fiscal Crisis in the Euro Zone. Lessons for Latin America.
- 11/25 **Adolfo Albo y Juan Luis Ordaz Díaz:** Migración mexicana altamente calificadaen EEUU y Transferencia de México a Estados Unidos a través del gasto en la educación de los migrantes.
- 11/26 **Adolfo Albo y Juan Luis Ordaz Díaz:** Highly qualified Mexican immigrants in the U.S. and transfer of resources to the U.S. through the education costs of Mexican migrants.
- 11/27 Adolfo Albo y Juan Luis Ordaz Díaz: Migración y Cambio Climático. El caso mexicano.
- 11/28 Adolfo Albo y Juan Luis Ordaz Díaz: Migration and Climate Change: The Mexican Case.
- 11/29 **Ángel de la Fuente y María Gundín:** Indicadores de desempeño educativo regional: metodología y resultados para los cursos 2005-06 a 2007-08.
- 11/30 **Juan Ramón García:** Desempleo juvenil en España: causas y soluciones.
- 11/31 Juan Ramón García: Youth unemployment in Spain: causes and solutions.
- 11/32 **Mónica Correa-López and Beatriz de Blas:** International transmission of medium-term technology cycles: Evidence from Spain as a recipient country.
- 11/33 Javier Alonso, Miguel Angel Caballero, Li Hui, María Claudia Llanes, David Tuesta, Yuwei Hu and Yun Cao: Potential outcomes of private pension developments in China.
- 11/34 Javier Alonso, Miguel Angel Caballero, Li Hui, María Claudia Llanes, David Tuesta, Yuwei Hu and Yun Cao: Posibles consecuencias de la evolución de las pensiones privadas en China.
- 11/35 **Enestor Dos Santos:** Brazil on the global finance map: an analysis of the development of the Brazilian capital market
- 11/36 **Enestor Dos Santos, Diego Torres y David Tuesta:** Una revisión de los avances en la inversión en infraestructura en Latinoamerica y el papel de los fondos de pensiones privados.
- 11/37 **Enestor Dos Santos, Diego Torres and David Tuesta:** A review of recent infrastructure investment in Latin America and the role of private pension funds.
- 11/ 38 **Zhigang Li and Minqin Wu:** Estimating the Incidences of the Recent Pension Reform in China: Evidence from 100,000 Manufacturers.
- 12/01 Marcos Dal Bianco, Máximo Camacho and Gabriel Pérez-Quiros: Short-run forecasting of the euro-dollar exchange rate with economic fundamentals.
- 12/02 **Guoying Deng, Zhigang Li and Guangliang Ye:** Mortgage Rate and the Choice of Mortgage Length: Quasi-experimental Evidence from Chinese Transaction-level Data.
- 12/03 **George Chouliarakis and Mónica Correa-López:** A Fair Wage Model of Unemployment with Inertia in Fairness Perceptions.
- 2/04 **Nathalie Aminian, K.C. Fung, Alicia García-Herrero, Francis NG:** Trade in services: East Asian and Latin American Experiences.
- 12/05 Javier Alonso, Miguel Angel Caballero, Li Hui, María Claudia Llanes, David Tuesta, Yuwei Hu and Yun Cao: Potential outcomes of private pension developments in China (Chinese Version).
- 12/06 **Alicia Garcia-Herrero, Yingyi Tsai and Xia Le:** RMB Internationalization: What is in for Taiwan?
- 12/07 **K.C. Fung, Alicia Garcia-Herrero, Mario Nigrinis Ospina:** Latin American Commodity Export Concentration: Is There a China Effect?
- 12/08 **Matt Ferchen, Alicia Garcia-Herrero and Mario Nigrinis:** Evaluating Latin America's Commodity Dependence on China.



- 12/09 **Zhigang Li, Xiaohua Yu, Yinchu Zeng and Rainer Holst:** Estimating transport costs and trade barriers in China: Direct evidence from Chinese agricultural traders.
- 12/10 **Maximo Camacho and Jaime Martinez-Martin:** Real-time forecasting US GDP from small-scale factor models.
- 12/11 J.E. Boscá, R. Doménech and J. Ferri: Fiscal Devaluations in EMU.
- 12/12 **Ángel de la Fuente and Rafael Doménech:** The financial impact of Spanish pension reform: A quick estimate.
- 12/13 Biliana Alexandrova-Kabadjova, Sara G. Castellanos Pascacio, Alma L. García-Almanza: The Adoption Process of Payment Cards -An Agent- Based Approach:
- 12/14 Biliana Alexandrova-Kabadjova, Sara G. Castellanos Pascacio, Alma L. García-Almanza: El proceso de adopción de tarjetas de pago: un enfoque basado en agentes.
- 12/15 Sara G. Castellanos, F. Javier Morales y Mariana A. Torán: Análisis del Uso de Servicios Financieros por Parte de las Empresas en México: ¿Qué nos dice el Censo Económico 2009?
- 12/16 **Sara G. Castellanos, F. Javier Morales y Mariana A. Torán:** Analysis of the Use of Financial Services by Companies in Mexico: What does the 2009 Economic Census tell us?
- 12/17 **R. Doménech:** Las Perspectivas de la Economía Española en 2012.
- 12/18 Chen Shiyuan, Zhou Yinggang: Revelation of the bond market (Chinese version).
- 12/19 **Zhouying Gang, Chen Shiyuan:** On the development strategy of the government bond market in China (Chinese version).
- 12/20 **Ángel de la Fuente and Rafael Doménech:** Educational Attainment in the OECD, 1960-2010.
- 12/21 **Ángel de la Fuente**: Series enlazadas de los principales agregados nacionales de la EPA, 1964-2009.
- 12/22 **Santiago Fernández de Lis and Alicia Garcia-Herrero:** Dynamic provisioning: a buffer rather than a countercyclical tool?
- 12/23 **Ángel de la Fuente:** El nuevo sistema de financiación de las Comunidades Autónomas de régimen común: un análisis crítico y datos homogéneos para 2009 y 2010.
- 12/24: **Beatriz Irene Balmaseda Perez, Lizbeth Necoechea Hasfield:** Metodología de estimación del número de clientes del Sistema Bancario en México.
- 12/25 Ángel de la Fuente: Series enlazadas de empleo y VAB para España, 1955-2010.
- 12/26 **Oscar Arce, José Manuel Campa y Ángel Gavilán:** Macroeconomic Adjustment under Loose Financing Conditions in the Construction Sector.
- 12/27 **Ángel de la Fuente:** Algunas propuestas para la reforma del sistema de financiación de las comunidades autónomas de régimen común.
- 12/28 **Amparo Castelló-Climent and Rafael Doménech:** Human Capital and Income Inequality: Some Facts and Some Puzzles.
- 12/29 **Mónica Correa-López y Rafael Doménech:** La Internacionalización de las Empresas Españolas.
- 12/30 Mónica Correa-López and Rafael Doménech: The Internationalisation of Spanish Firms.
- 12/31 **Robert Holzmann, Richard Hinz and David Tuestah:** Early Lessons from Country Experience with Matching Contribution Schemes for Pensions.

The analysis, opinions, and conclusions included in this document are the property of the author of the report and are not necessarily property of the BBVA Group.

 ${\tt BBVA \ Research's \ publications \ can \ be \ viewed \ on \ the \ following \ website: \ http://www.bbvaresearch.com}$

Contact details

BBVA Research

Paseo Castellana, 81 - 7th floor 28046 Madrid (Spain)

Tel.: +34 91 374 60 00 and +34 91 537 70 00 Fax: +34 91 374 30 25 bbvaresearch@bbva.com www.bbvaresearch.com